Author's response to reviews

Title: Prevalence and predictors of antenatal depression among Chinese women in their third trimester: a cross-sectional survey

Authors:

Yingchun Zeng ZYC (chloezengyc@hotmail.co.uk)
Ying Cui CY (cui_ying@outlook.com)
Jie Li LJ (jie_li@126.com)

Version: 2 Date: 23 December 2014

Author's response to reviews: see over
Cover letter

Dear Editor-in-Chief,

I am writing to submit a manuscript entitled “Prevalence and predictors of antenatal depression among Chinese women in their third trimester: a cross-sectional survey” for publication in *BMC Psychiatry*. This manuscript has not been published previously and is not under consideration for publication elsewhere. The authors have no conflicts of interest to disclose. This study provides a cross-sectional snapshot of the prevalence and relevant predictors of antenatal depression among Chinese women in their third trimester. Implications to clinical practice, research and education were discussed accordingly.

Relevant revisions were added accordingly:
1. Page and line numbers were added in this revised manuscript;
2. Acknowledgement was added accordingly in this revised manuscript;
3. This study has no sources of fund;
4. This cross-sectional study has obtained formal approval from an ethics committee of the studied hospital.

Thank you for kind considerations!

Yours sincerely,

Yingchun Zeng, RN, MPhil
Research Associate Professor
Department of Obstetrics and Gynecology
The Third Affiliated Hospital of Guangzhou Medical University
Guangzhou
China 510150